Author’s response to reviews

Title: Human papillomavirus in semen and the risk for male infertility: a systematic review and meta-analysis

Authors:

Zhangyan Lyu (lvzy2016@126.com)
Xiaoshuang Feng (fengxs2016@163.com)
Ni Li (nli@cicams.ac.cn)
Wei Zhao (zhaowei0714@nicemice.cn)
Luopei Wei (weiluopeiwlp@163.com)
Yuheng Chen (yhchen@cicams.ac.cn)
Wenjing Yang (wenjing86824@126.com)
Hongxia Ma (hongxiama@njmu.edu.cn)
Bing Yao (yaobing@nju.edu.cn)
Kai Zhang (kai87968@sina.com)
Zhibin Hu (zhibin_hu@njmu.edu.cn)
Hongbing Shen (hbshen@njmu.edu.cn)
Dong Hang (hangdong@njmu.edu.cn)
Min Dai (daimin2002@hotmail.com)

Version: 3 Date: 26 Sep 2017

Author’s response to reviews:

Dear Editor,

Thank you very much for your decision letter. The manuscript has been revised accordingly, with our point-by-point responses attached below.
Yours sincerely,

Dong Hang, Ph.D.

Comments from Editor:

1. Figure files should contain only the image/graphic, as well as any associated keys/annotations. If titles/legends are present within the figure files, please remove them.

Response: We have removed the titles and legends which were presented within the figure files.

2. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: We have upload the manuscript, figures and additional files according to the requirements.